Author's response to reviews

Title: Acute appendicitis caused by endometriosis: A case report

Authors:

Styliani Laskou (stilianl@live.com)
Theodossis S Papavramidis (papavramidis@hotmail.com)
Angeliki Cheva (antacheva@yahoo.gr)
Nick Michalopoulou (nickos.michalopoulou@gmail.com)
Charilaos Koumouris (charilaoskoumouris@gmail.com)
Isaak Kesisoglou (ikesis@hotmail.com)
Spiros Papavramidis (spapavra@med.auth.gr)

Version: 2 Date: 17 November 2010

Author's response to reviews: see over
Dear Editor,
Thank you for your time. We thank both reviewers for their comments. All changes were made as suggested. All changes were tracked down by the track changes tool of Word.

Reviewer: Dr. Marinis

Q1) Abstract: Introduction: 3rd sentence: please rephrase the following: "... accounts of its non typical manifestations are incomplete", because it is not well understood.
R1) Done as suggested

Q2) Case presentation: 2nd sentence: please change "iliac fossa" to "lower quadrant abdominal pain".
R2) Done as suggested.

Q3) Case presentation: 3rd sentence: please delete "tenderness" and add the word "abdominal" to "guarding and rigidity".
R3) Done as suggested

Q4) Total word count of the manuscript should be max 2000 words and references max 15. Please check.
R4) The word count for the manuscript is 1573. References are more than 15, however, the only scientific merit of one author is to be cited appropriately.

Reviewer: Dr. Milias

Q1. Too extensive text in "Discussion". There is no need for such explanation of the different theories of endometriosis.
R1. Indeed the discussion was shortened by altering the above-mentioned paragraph.

Q2. Bad quality of English in several occasions. The authors seem to translate greek in english and not write in english (e.g. pathologic assessments, "she ahd no fever, But nausea..., McBurney's point)
R2: Major linguistic corrections made at several points